Author's response to reviews

Title: Incidence of chemotherapy-induced amenorrhea associated with epirubicin, docetaxel and navelbine in younger breast cancer patients

Authors:

Wen-Bin Zhou (zhouwenbin_1984@qq.com)
Hong Yin (77077977@qq.com)
Xiao-An Liu (liuxiapaon@126.com)
Lin Chen (hxchenlin@yahoo.com.cn)
Jun-Cheng Dai (stata9.2@163.com)
Ai-di Tao (enbens@njimu.edu.cn)
Ling Chen (rainbowlyn@gmail.com)
Jing-Jing Ma (honeymij@hotmail.com)
Li-Jun Ling (pangpangzheng@sina.com)
Shui Wang (ws0801@hotmail.com)

Version: 4 Date: 22 May 2010

Author's response to reviews: see over
Dear editor:

Thank you very much for your letter dated on May 21, 2010 in which you sent us the Referees’ Reports on our manuscript (MS: 8378301913424560). We would also like to thank the referees and editor for their valuable comments and suggestions.

The concerns raised by the referees are responded as follows. We also added the email address in page 3. Updated reference 21 is to include full details. In addition, a number of minor errors are corrected.

I hope the manuscript has been improved satisfactorily and that it will be accepted for publication in your journal.

With best wishes.

Zhou Wenbin, MD.
Department of Breast Surgery
the First Affiliated Hospital of Nanjing Medical University
Email: zhouwenbin_1984@foxmail.com
Response to Reviewers and editor

Referee Mikkel Rosendahl:
In their revised manuscript, the authors have presented a clear and interesting paper to which I only have a few comments.
Response: We are grateful for the positive comments on our manuscript by the reviewer and have revised our manuscript according to reviewer’s suggestion point by point. The changes in our revised manuscript were marked in red.

Minor Essential Revisions:
Abstract, conclusion; "... whereas doco and navel were not associated with higher rates of CIA" Higher than what?
Response: We are sorry for not describing our conclusion clearly. We are greatly appreciated for the reviewer’s suggestion. We modified it in line 3-4, Page 4.